Author’s response to reviews

**Title:** The natural history of, and risk factors for, progressive Chronic Kidney Disease (CKD): the Renal Impairment In Secondary care (RIISC) study; rationale and protocol

**Authors:**

Stephanie Stringer (stephanie.stringer@uhb.nhs.uk)
Praveen Sharma (praveen.sharma@nhs.net)
Mary Dutton (mary.dutton@uhb.nhs.uk)
Mark Jesky (mark.jesky@uhb.nhs.uk)
Khai Ng (KhaiPing.Ng@u hb.nhs.uk)
Okdeep Kaur (okdeep.kaur@u hb.nhs.uk)
Iain Chapple (I.L.C.CHAPPLE@bham.ac.uk)
Thomas Dietrich (t.dietrich@bham.ac.uk)
Charles Ferro (charles.ferro@u hb.nhs.uk)
Paul Cockwell (paul.cockwell@u hb.nhs.uk)

**Version:** 2  **Date:** 15 January 2013

**Author’s response to reviews:**

1. Lead/corresponding author changed
2. Lead author email address changed
3. Tables added
4. Documents relating to ethics/R&D approval removed